Author's response to reviews

Title: Assessment of soy phytoestrogens' effects on bone turnover indicators in menopausal women with osteopenia in Iran: a before and after clinical trial

Authors:

Arezoo Haghighian Roudsari (ahaghighian@yahoo.com)
Farideh Tahbaz (faridehta@yahoo.com)
Arash Hossein-nezhad (arash_hsi@yahoo.com)
Bahram Arjmandi (arjmand@okstate.edu)
Bagher Larijani (emrc@sina.tums.ac.ir)
Seyed Masoud Kimiagar (smkimiagar@yahoo.com)

Version: 4 Date: 20 October 2005

Author's response to reviews:

Dear Dr Gebrayel,

The requested formatting changes (including full affiliation and removing the highlights) were made and the revised version of article was uploaded in submission system.

Thank You,
Bagher Larijani, MD